Author’s response to reviews

Title: Strategic approaches to enhanced health service delivery for Aboriginal and Torres Strait Islander people with chronic illness: A qualitative study

Authors:

   Clive Aspin (clive.aspin@sydney.edu.au)
   Ngiare Brown (ngiare@aida.org.au)
   Tanisha Jowsey (tanisha.jowsey@anu.edu.au)
   Laurann Yen (laurann.yen@anu.edu.au)
   Stephen Leeder (stephen.leeder@sydney.edu.au)

Version: 2 Date: 22 September 2011

Author’s response to reviews:

I have amended the manuscript as suggested. I have added a citation that refers to a previous paper. I have also added a comment explaining how that paper complements this paper. Both papers draw on the same data from the same sample of participants but each paper has a distinctly different focus. The former examines self-management issues and this one looks at health service delivery.